Author's response to reviews

Title: Inter-population variations in concentrations, determinants of and correlations between 2,2',4,4',5,5'-hexachlorobiphenyl (CB-153) and 1,1-dichloro-2,2-bis (p-chlorophenyl)-ethylene (p,p'-DDE): a cross-sectional study of 3161 men and women from Inuit and European populations

Authors:

Bo AG Jonsson (Bo_A.Jonsson@med.lu.se)
Lars Rylander (lars.rylander@med.lu.se)
Christian Lindh (christian.lindh@med.lu.se)
Anna Rignell-Hydbom (anna.rignell-hydbom@med.lu.se)
Aleksander Giwercman (aleksander.giwercman@med.lu.se)
Gunnar Toft (gutof@as.aaa.dk)
Henning S Pedersen (hsp@gh.gl)
Jan K Ludwicki (k.ludwicki@pzh.gov.pl)
Katarzyna Goralczyk (k.goralczyk@pzh.gov.pl)
Valentyna Zvyezday (dimusic@ic.kharkov.ua)
Marcello Spano (spanomrc@casaccia.enea.it)
Davide Bizzaro (d.bizzaro@univpm.it)
Eva C Bonefeld-Jorgensen (EBJ@MIL.AU.DK)
Gian-Carlo Manicardi (manicardi.giancarlo@unimore.it)
Jens Peter Bonde (JPBON@as.aaa.dk)
Lars Hagmar (lars.hagmar@med.lu.se)

Version: 5 Date: 2 November 2005

Author's response to reviews:

Dear Dr Ozonoff,

Thank you for accepting our manuscript. With respect to the remaining technical issues:

Numbers need to be removed from the left-hand column.

Done

Abstract needs to be 350 words or less.
Done (exactly 350)

List of abbreviations is needed. (see instructions for authors regarding placement in the text)
Done, but I still think your instructions somewhat ambiguous. I cite from your instructions:
"If abbreviations are used in the text either they should be defined in the text where first used, or a list of abbreviations can be provided, which should precede the competing interests and authors' contributions."

However, your practice seems to be "both and" rather than "either or". I would appreciate your comment on this.

Entire paper needs to be unjustified on the right, including references.
Done

Please recheck references to make sure all of them follow EH instructions. Most do, but here and there I can see errors.
Done (as far as I have understood the instructions)

Tables need to follow EH instructions (http://www.ehjournal.net/info/instructions/). There appear to be some cells that do not have an outline that need it. Please also make sure that tables fall within the margins. Please specify which "cells that do not have an outline that need it". I cannot find any. Moreover, I can check in the pdf-file that I see the tables (which I do), but I see nowhere in your instructions a definition of margin
INUENDO is capitalized in some places and not in others - why is this?

Revised

Title page needs an affiliation and email address.
I don't really understand this comment. There are affiliations and email addresses for all co-authors. What we earlier have discussed is whether the virtual organisation Inuendo could have an email address or not. In your previous letter you wrote: "The web address issue will have to be checked with BMC.", but you did not comment on this is your last letter.

Anyhow, I submit a new version and look forward to your response.

Best regards,

Lars Hagmar